Author's response to reviews

Title: Dose and polymorphic genes xrcc1, xrcc3, gst play a role in the risk of developing erythema in breast cancer patients following single shot partial breast irradiation after conservative surgery

Authors:

Elisabetta Falvo (falvo@ifo.it)
Lidia Strigari (strigari@ifo.it)
Gennaro Citro (citro@ifo.it)
Carolina Giordano (carolinagiordano1@gmail.com)
Stefano Arcangeli (stefano.arcangeli@yahoo.it)
Antonella Soriani (soriani@ifo.it)
Daniela D’Alessio (dany.dalessio@libero.it)
Paola Muti (dirscire@ifo.it)
Giovanni Blandino (blandino@ifo.it)
Isabella Sperduti (sperduti@ifo.it)
Paola Pinnarò (paola.pinnaro@gmail.com)

Version: 5 Date: 11 July 2011

Author's response to reviews:

Dear Editor, the text has been duly modified according to your suggestions.
Sincerely
Elisabetta Falvo